Author’s response to reviews

Title: Drama as a pedagogical tool for practising death notification - experiences from Swedish medical students

Authors:

Anna Nordström (anna.nordstrom@idrott.umu.se)
Annncristine Fjellman-Wiklund (annncristine.fjellman-wiklund@physiother.umu.se)
Tomas Grysell (tomas.grysell@uadm.uu.se)

Version: 4 Date: 29 August 2011

Author’s response to reviews:

Dear editor,

We have now made the requested changes:
A new abstract has been uploaded, identical to the abstract in the manuscript.
Highlighting/tracking has been removed from the manuscript.
Vertical lines have been removed from the tables.